Author's response to reviews

Title: Factors associated with seizure occurrence and long-term seizure control in Pediatric brain arteriovenous malformation: a retrospective analysis of 89 patients

Authors:

Shuang Liu (liushuang3302@163.com)
Hong-xu Chen (shenwaichx@163.com)
Qing Mao (gingmao2000@yahoo.com.cn)
Chao You (youchao1006@sina.cn)
Jian-guo Xu (jianguo_1229@sina.com)

Version: 9 Date: 16 July 2015

Author's response to reviews:

Dear Mr Ines Lopes,

Sincerest thanks for your response and Editors comments on our manuscript. According to your request, we response to the concerns point counter point. Firstly, we confirm that the ethical approval for this study, including retrospective data collection, was granted by the Biological and Medical Ethics Committee of West China Hospital. This information could be found in the methods section (page 6, line 21 -22 and page 7, line 1). Secondly, we provide a full postal address for the corresponding author (Page 1, lines 8-9). Furthermore, we have reedited the manuscript to conform to the journal style. Thank you for your advice again.

Best Regards.

Yours Sincerely,

Liu Shuang

Affiliation: Department of Neurosurgery, West China hospital of Sichuan University, 37 Guoxue Street, Chengdu 610041, People's Republic of China

E-mail: liushuang3302@163.com